Author's response to reviews

Title: Severe gastric variceal haemorrhage due to splenic artery thrombosis and consecutive arterial bypass

Authors:

Klaus T von Trotha (kvontrotha@ukaachen.de)
Marcel Binnebösel (mbinneboesel@ukaachen.de)
Son Truong (struong@ukaachen.de)
Florian F Behrendt (fbehrendt@ukaachen.de)
Hermann E Wasmuth (hwasmuth@ukaachen.de)
Ulf P Neumann (unueumann@ukaachen.de)
Marc Jansen (mjansen@ukaachen.de)

Version: 8 Date: 17 June 2011

Author's response to reviews:

Ref.: 1938887921447171

Severe gastric variceal haemorrhage due to splenic artery thrombosis and consecutive arterial bypass

Dear Rachel Neilan,

Thank you very much for the quick reply. The formatting changes have been made as you suggested.

For some unknown reason the submission system wouldn't let me choose the correct journals "Gastroenterology" and "Surgery". I kindly ask this to be changed.

Sincerely yours,

Klaus-Thilo von Trotha

Correspondence to:
Klaus-Thilo von Trotha, MD
Department of Surgery, RWTH Aachen University Hospital
Pauwelsstr. 30
52074 Aachen, Germany
Phone: +49-241-80-89500
Fax: +49-241-80-82417
Email: kvontrotha@ukaachen.de